

| | Reviewer Comments (Each reviewer should contribiute to and sign one form per project.) | | | | | | | | | | | | | |
|---------|--|----------|----------|----------|-----------------|-----------|-----------|---------|-----------|-----------|--------------|----------|------------|----|
| Name o | of Revie | wer: | | | | | | | | | | | | |
| Name o | of Princi | ipal Inv | estigato | or(s): | | | | | | | | | | |
| Project | : Title: | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| Did the | | | | | | | | | | | /ER for list | | Yes | No |
| | | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 | 11 | 12 | |
| Do you | recomr | nend t | hat this | applica | tion (se | lect one) |): | | | | | | | |
| • | | | | JT any m | | | | | | | | | | |
| | | | | • | | | ations ar | e addre | ssed (IRE | B Chair v | vill incorp | orate re | evisions.) | |
| | | | | e COMN | | | | | , | | · | | , | |
| | | • • | • | | | • | | | | | | | | |
| сомм | ENTS: | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | |
| | | | | | | | | | | _ | | | | |
| | | Signat | ture of | IRB Com | mittee I | Member | | | | | | Da | ite | |
| | | | | | | | | | | | | | | |
| | | Signat | uro of ! | DR Com | mittee N | | | | | | | | | |
| | | Signat | ure of I | RB Comr | milee N | nember | | | | | | υa | ite | |

Informed Consent Checklist

U.S. Department of Health & Human Services http://www.hhs.gov/ohrp/policy/consentckls.html

- 1. A statement that the study involves research
- 2. An explanation of the purposes of the research
- 3. The expected duration of the subject's participation
- 4. A description of the procedures to be followed
- 5. Identification of any procedures which are experimental
- 6. A description of any reasonably foreseeable risks or discomforts to the subject
- 7. A description of any benefits to the subject or to others which may reasonably be expected from the research
- 8. A disclosure of appropriate alternative procedures or courses of treatment, if any, that might be advantageous to the subject
- 9. A statement describing the extent, if any, to which confidentiality of records identifying the subject will be maintained
- 10. For research involving more than minimal risk, an explanation as to whether any compensation, and an explanation as to whether any medical treatments are available, if injury occurs and, if so, what they consist of, or where further information may be obtained
- 11. Research, Rights or Injury: An explanation of whom to contact for answers to pertinent questions about the research and research subjects' rights, and whom to contact in the event of a research-related injury to the subject
- 12. A statement that participation is voluntary, refusal to participate will involve no penalty or loss of benefits to which the subject is otherwise entitled, and the subject may discontinue participation at any time without penalty or loss of benefits, to which the subject is otherwise entitled